



Article The Impact of the Digital Economy on Supply Chain Security: Evidence from China's Wooden Furniture Industry

Yiyi Luo ¹, Yilin Chen ¹, Chenlu Tao ², Chao Yang ¹, Futao Xiang ³, Chang Xu ³, and Fanli Lin ^{3,*}

- ¹ School of Economics and Management, Beijing Forestry University, No. 35 Tsinghua East Road, Beijing 100083, China; luoyiyi0127@bjfu.edu.cn (Y.L.); elaainee0912@163.com (Y.C.); taochenlu@ncepu.edu.cn (C.Y.)
- ² School of Economics and Management, North China Electric Power University, No. 2 Beinong Road, Beijing 100096, China; yangchao99@163.com
- ³ School of Finance and Public Administration, Anhui University of Finance and Economics, No. 962 Caoshan Road, Bengbu 233030, China; xiangfutao@aufe.edu.cn (F.X.); xuchang@aufe.edu.cn (C.X.)
- * Correspondence: linfanli@aufe.edu.cn

Abstract: Supply chain security is a major prerequisite for China's successful industrial modernization, while the digital economy has significantly contributed to industrial transformation and upgrading. This study considers China's wooden furniture industry as its research object, constructing an evaluation index system of the digital economy and supply chain security of the wooden furniture industry. Then, it studies the impact of the digital economy on supply chain security through theoretical analysis and empirical methods using the two-way fixed model of provinces and time. The findings demonstrate that the digital economy effectively enhances the level of supply chain security in China's wooden furniture industry, further validating the digital economy's positive externality impact on the traditional real economy. The impact mechanism test shows that inventory turnover capacity is the focal point for the digital economy to improve the supply chain security of the wood furniture industry, specifying the starting point for that industry's digital transformation. The heterogeneity findings show that the role of the digital economy in improving the wood furniture industry's level of supply chain security is more significant in inland areas than in coastal areas. Additional analyses found a threshold effect of the digital economy's impact on supply chain security, indicating its limitations. This study explores the impact of the digital economy on the real economy from a traditional manufacturing industry, enriching research on the positive externalities of the digital economy as well as providing a reference for traditional manufacturing industries, such as that of wooden furniture, to probe the embedding points of the digital economy and appropriate digital transformation.

Keywords: digital economy; supply chain security; wooden furniture industry; mediation effect; threshold effect

1. Introduction

Supply chain security is essential to promoting the high-quality development of national industries, ensuring smooth operation of the real economy, and building new development patterns, as well as maintaining national economic security as the cornerstone [1]. However, with the anti-globalization and trade protectionism trend, China's supply chain security has been greatly threatened [2,3], such that unblocking the supply chain security blockages caused by various developed countries in the process of desinicization is urgent. Digital technology offers application value in the process of social and economic development, as well as being a driver of China's breakthrough industrial transformation and upgrading, together with high-quality economic growth [4]. Therefore, an in-depth analysis of the impact of China's digital economy on supply chain security and specifically the practical problem of digital and real economy integration is of practical significance



Citation: Luo, Y.; Chen, Y.; Tao, C.; Yang, C.; Xiang, F.; Xu, C.; Lin, F. The Impact of the Digital Economy on Supply Chain Security: Evidence from China's Wooden Furniture Industry. *Forests* **2024**, *15*, 879. https://doi.org/ 10.3390/f15050879

Academic Editors: Andreja Pirc Barčić, Martina Basarac Sertić, Daniela Ventsislavova Georgieva and Matej Jošt

Received: 9 April 2024 Revised: 11 May 2024 Accepted: 15 May 2024 Published: 17 May 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). for China to cope with global supply chain risks, such as chain-breaking, short-chaining, and regionalization. The wooden furniture industry is a typical traditional manufacturing industry in China with status in the international trade market. According to the United Nations (UN) Commodity Trade Statistics Database, in 2021 the global furniture gross output value was USD 490 billion, with China accounting for 37%; while global furniture consumption was USD 477 billion (China accounted for 28%); and global furniture exports were USD 160 billion (China accounted for 35%). China has become the world's largest producer, exporter, and consumer of wooden furniture. Despite the industry's large volume, there remains supply chain network confusion and supply channel issues [5]. There are also real problems of high dependence on external sources in the production process of wooden furniture, such as automation and digital machinery, and the supply of raw materials for the production of logs, sawn timber, hardware, paints, and glues [6].

To explore the impact of the digital economy on the supply chain security of China's wooden furniture industry, analyze the problems and blockages, and identify the inherent influence mechanism, this study constructs an evaluation index system for the digital economy and supply chain security using a comparison method. The digital economy mechanism affecting the supply chain security of the wooden furniture industry is revealed through the mediating effect model, identifying a specific direction for deep integration between the digital economy and traditional manufacturing industry. In addition, regional heterogeneity in the digital economy's impact was explored from the perspective of economic geography. The reason for such heterogeneity was also probed using the threshold effect test.

This article aims to explore ways to enhance the security of the furniture supply chain, analyze the impact of the digital economy on its supply chain security, and to some extent enrich and expand the research scope of the impact of the digital economy on the traditional manufacturing industry. Specifically, this study contributes to the literature in the following two aspects. First, through quantitative analysis, it verifies the effectiveness of the digital economy in enhancing the supply chain security of China's wooden furniture industry, which provides evidence for the integration of the digital economy with the real economy, thus optimizing supply chain security. Compared to existing research results on the modernized manufacturing industry, this study discusses the impact of the digital economy on the supply chain security of a traditional industry, offering a further exploration and supplement to its digital transformation and upgrading. Second, it identifies the limitations of the digital economy in improving the supply chain security of the traditional manufacturing industry and provides empirical evidence for the appropriateness of digital transformation in such industries. Compared with high-tech manufacturing, selecting proper digital embedding points and an adequate digital transformation path are more likely to improve the level of supply chain security in the industry. At the same time, these findings have practical significance for the digital transformation and upgrading of the traditional manufacturing industry, optimizing the supply structure, alleviating the "bullwhip effect" to improve supply chain efficiency, and improving the security of other traditional manufacturing supply chains.

The following sections are organized as follows: The Section 2 provides a comprehensive literature review, summarizing relevant research on the intersection of the digital economy and supply chain in the context of the wooden furniture industry. The Section 3 focuses on the development and implementation of a supply chain security evaluation index system, highlighting the integration of supply chain security and digital economy metrics. The Section 4 outlines the theoretical framework and introduces two key research hypotheses central to this study. The Section 5 delves into the research design, detailing methodology, data selection, and sources. Section 6 presents analysis, discussion, and avenues for further research based on empirical findings. Lastly, Section 7 offers policy recommendations informed by the conclusions drawn in this paper.

2. Literature Review

The digital economy has dramatically driven economic growth [7]. A careful review of relevant literature on the digital economy shows that scholarly attention to the digital economy's conceptual connotations has gradually shifted from initial topics, such as ecommerce and the digital divide [8,9], to informatization, digitization, innovation R&D, platform economy, big data, and artificial intelligence, among others [10,11], with an indepth discussion of the positive externalities brought about by the digital economy [12]. For instance, some scholars note that the digitization of the customer base and knowledge spillovers are effective in enhancing suppliers' innovation capabilities [13], while others believe that the continuous optimization of human resource structures by means of digital technology is a priority for enhancing enterprises' innovation capacity [14]. Studies have also discussed the impact of the digital economy on the transformation of the division of labor in the value chain from a more macroscopic point of view, arguing that digital transformation effectively improves the resilience of supply chains in various industries [15]. Measuring the digital economy development level is also a research hotspot in this field. Current studies mainly consider two aspects. The first accounts for the size of the digital economy through input-output tables, growth accounting frameworks, and the construction of satellite accounts [16], whose basic logic lies in stripping away the economic growth achieved through digital forms in the national economy and evaluating the degree of digitalization development in different industries through measuring the value-added of the digital economy [15]. The second aspect uses an evaluation index system constructed through the comparative method to assess the development level of the digital economy by integrating evaluation indices directly related to digital technology [17]. For example, Xu et al. [18] portrayed the level of development of the digital economy in Chinese provinces (cities and districts) with the help of a digital financial inclusion index, while Serna et al. [19] constructed a system of evaluative indicators through textual analysis and the construction level of relevant infrastructure to assess the progress of Latin American higher education institutions in digital transformation.

In the latest round of the scientific and technological revolution, modernization of the supply chain is an inevitable trend along with industrial change, and the stability of supply chain security is an inevitable requirement for national modernization [20,21]. Specific research has focused on current supply chain problems and methods to improve supply chain security. Dong [22] identified the lack of basic capabilities and innovation capacity as important constraints on supply chain security in China. The argument that although China's supply chain security problems have always existed, they have been resolved at different stages of history is also being made. Li and Zhao [23] reviewed China's economic achievements at various stages since the People's Republic's (PRC) founding and the strategies adopted to secure its supply chain, noting that China's supply chain security has benefited from the government's ability to change the path of industrial modernization from exogenous to endogenous based on national conditions [24]. These changes include the reform of the economic system and continuous opening to the outside world to stabilize and optimize supply chain security. Regarding improving the level of supply chain security, a supply chain modernization logic-based analysis by Yang [25] found that to increase supply chain resilience, the most important direction is to optimize the supply chain's knowledge configuration and then shape a more durable, competitive, and innovative modern supply chain.

High innovation, strong permeability, and wide coverage are the significant features and advantages of the digital economy [26], which practically contribute to improving supply chain security; thus, exploring how the digital economy can do so is also a hot research topic. Globally, Yang et al. [27] discussed the impact of the digital economy on global supply chain security, concluding that building a platform economy and a virtual digital image of the supply chain network through the Internet can effectively promote more small- and medium-sized enterprises (SMEs) to participate in inclusive trade globally, thus improving supply chain security. Other perspectives also suggest that data resources are effective in enhancing the added value of products and services and bringing growth effects to corresponding markets [28]; therefore, data are the foundation for the digital economy's impact on supply chain security [29]. In addition, Peng and Zhou [30] argued that information sharing and information transfer efficiency are crucial for the digital economy to improve supply chain security. From an internal governance perspective, digitalization enhances enterprises' risk-taking capacity [31], reduces business management costs, and ensures supply chain security and stability [32]. The digital economy's impact on the supply chain security of China's wooden furniture industry garnered fewer research results, with the focus on problems existing in the wooden furniture industry itself and the research scope being dominated by the development of the local wooden furniture industry, rarely from the supply chain perspective [33,34].

In summary, the positive impact of the digital economy on society and the economy has been fully affirmed in the existing literature, and measuring the digital economy's development has also been discussed from multiple perspectives and dimensions. The significant body of research on supply chain security provides reference value for this study regarding research concepts and methodologies. While the topic is exploratory based on the needs of the times, the research object mostly focuses on the high-tech industry, and there remains a gap for further research in the traditional manufacturing industry. The wooden furniture industry, as a typical representative of China's traditional manufacturing industry, has been the subject of qualitative research but little quantitative research. Therefore, this paper can expand academic research to a certain extent.

3. Constructing and Accounting a Supply Chain Security Evaluation Index System for the Wooden Furniture Industry

3.1. Constructing Supply Chain Security Evaluation Indicators for the Wooden Furniture Industry

Supply chain security can be measured from two primary aspects: (1) Considering possible supply chain risks and summarizing factors that negatively affect supply chain security and (2) exploring the ways in which supply chain security can be better safeguarded and providing solutions to supply chain crises. This study considers the basic needs for developing the wooden furniture industry as a starting point and constructs an evaluation index system for wooden furniture supply security from four supply perspectives: capital, labor, production technology, and raw material supply.

Regarding capital supply, this study takes the total assets and gearing ratio of the furniture manufacturing industry as the reference indices for the level of capital supply. This is because, first, the furniture manufacturing industry's total assets are a more intuitive reflection of China's provinces (cities and districts) in the overall development of the wood furniture industry. In general, having more total assets is more conducive to the sustainable development of an industry [35]. Second, from the view of ease of access to capital, although enterprises' gearing ratio is not uniformly higher or lower, the more favorable or unfavorable it is to the development of the enterprise, the relatively higher the gearing ratio reflects the degree of social capital to the development of the wood furniture industry [36]. Moreover, when an enterprise has a strong solvency capacity, it tends to obtain financial support through financing loans [37]. Therefore, the total assets and gearing ratio of the wood furniture manufacturing industry were used as the evaluation indices of capital supply in the wood furniture industry.

Next, the employment rate is mostly used in the existing literature to reflect labor supply [38]. However, because the wooden furniture industry is a traditional, labor-intensive industry, the income level of those engaged in furniture manufacturing is relatively low and unattractive to those already employed [39]. Thus, it would be more reasonable to use the number of under-employed people as an indicator of labor supply.

Regarding production technology supply, this indicator mainly refers to the supply of wood furniture production and processing equipment. A lower import dependence on such equipment indicates a higher level of supply chain security in the industry [40], and the greater the number of importing source countries, the higher the security of the supply chain [41]. Moreover, higher exports of wood furniture processing equipment produced in China reflect higher self-sufficiency in furniture production equipment. Thus, the import and export values of wood furniture production and processing equipment as well as import diversification are used as indicators to examine production technology supply [42].

Finally, raw material supply is determined by forest resource endowment, timber transshipment conditions, and raw materials supply. First, forest resource endowment is a basic condition for the stable and sustainable development of the wood processing industry [43]. Therefore, this study selected provinces (cities and districts) as the volume of live wood storage, and the total production of logs and sawn timber as the index of forest resource endowment. Second, timber transshipment is key to solving the "last mile" problem of raw materials from the source to wood furniture producers, hence the gross value of timber harvesting as an indicator [44]. In addition, current research indicates that raw material supply is supplemented by the production of manmade boards and the import of logs and sawn timber [45]. The details are presented in Table 1.

Table 1. Evaluation index system of supply chain security in China's wooden furniture industry.

Primary Indicator	Secondary Indicator	Data Source	Mean	Std.	Character
Capital supply	Log of total assets of the furniture manufacturing industry	China Furniture Yearbook	3.08	2.69	+
	Log of gearing ratio	China Economic Census	3.55	1.91	+
Labor supply	Log of urban registered unemployed population	National Bureau of Statistics	3.04	0.72	+
	Log of imports of wood furniture production and processing equipment	Statistical Yearbook of China's Light Industry	-2.74	2.56	-
Production technology supply	Log of exports of wood furniture production and processing equipment	Statistical Yearbook of China's Light Industry	3.08 2.69 3.55 1.97 3.04 0.77 -2.74 2.56 16.52 2.80 2.11 1.03 9.85 1.77 5.33 1.66 1.87 1.87 5.09 2.10 5.66 5.13	2.80	+
	Diversification of processing equipment imports	Database of the General Administration of Customs of China	2.11	1.05	+
	Log of standing tree stock	National Bureau of Statistics	9.85	1.71	+
	Log of total log sawn timber production	Database of the General Administration of Customs of China	5.33	1.63	+
	Log of gross value of timber harvesting and transportation	National Bureau of Statistics	1.87	1.87	+
Raw material supply	Log of wood-based panel production	Database of the General Administration of Customs of China	neral	2.16	+
	Log of imports of sawn timber	Database of the General Administration of Customs of China	5.66	5.13	-
	Log of total imports of wood-based panels	Database of the General Administration of Customs of China	-3.09	3.32	+

Note: "-" indicates that this indicator has a negative impact in the entire indicator system, while "+" indicates that this indicator has a positive impact in the entire evaluation indicator system.

3.2. Constructing an Evaluation Index System for the Digital Economy

This study aims to explore the impact of the digital economy on China's wood furniture industry's supply chain security and analyze its intrinsic impact mechanism and the heterogeneity of the impact. Hence, it is necessary to assess the level of digital economic development in Chinese provinces (cities and districts) accurately as a core explanatory variable. Given data accessibility and completeness, this study adopted a comparative approach to assess the level of digital economic development, referring to Pei and Yun's [46] selection method for digital economy evaluation indices to construct China's provincial digital economy development evaluation index. The index system has three dimensions: Digital industry activity, digital innovation activity, and digital user activity. These dimensions were selected as follows. First, digital industry activity effectively reflects the degree of social and economic utilization of digital technology, which is the material basis for the development of the digital economy. Second, digital innovation activity is critical to reflecting the digital user activity further refines the application of digital technology at the micro level, fully mirroring the completeness of the digital economy's foundation. The details are presented in Table 2.

Table 2. Evaluation index system of digital economy development level.

Primary Indicator	Secondary Indicator	Data Source	Mean	Std.	Character
	Log of main business income of high-tech enterprises	China Economic Census	6.76	1.90	+
Digital industry activity	Employed in the information transmission, software and information technology services industry	China Statistical Yearbook	1.66	1.01	+
	Density of Internet broadband access ports	China Statistical Yearbook	3.09	2.69	+
	Mobile telephone exchange capacity	China Statistical Yearbook	8.21	1.02	+
	R&D full-time equivalents (10,000 people per year)	China Statistical Yearbook	10.31	13.15	+
Digital innovation	Log of R&D internal expenditures	China Statistical Yearbook	4.95	1.57	+
activity	Log of domestic patent applications granted	China Statistical Yearbook	9.38	1.75	+
	Log of technology market turnover	China Statistical Yearbook	4.09	1.95	+
	Mobile telephones per 100 people	China Statistical Yearbook	4.18	0.62	+
Digital	Log of the number of Internet users	China Statistical Yearbook	7.11	1.25	+
user activity	Log of total telecommunication business	China Statistical Yearbook	4.02	1.09	+
	Log of the number of Internet access services places	China Statistical Yearbook	7.99	0.90	+

Note: "+" indicates that this indicator has a positive impact in the entire evaluation indicator system.

After processing the above indicators, the entropy weight method was used to measure the supply chain security index of the wooden furniture industry and the evaluation index of digital economy development. The contribution and uncertainty of the attributes are quantified by calculating the weights and entropy values of each indicator. In the calculation process, the decision matrix is first standardized, and then the attribute weight and entropy value are calculated by using the entropy value formula, and then the contribution degree of the attribute is obtained through normalization and multiplication operations.

4. Theoretical Basis and Research Hypotheses

4.1. Theoretical Basis

Chain theory is an essential cornerstone for studying supply chain security issues. The realization and creation of value are the fundamental logic of supply chain formation, while the social division of labor is its specific performance. Thus, the supply chain is continuously extended under the constant action of the trading mechanism [47]. Supply chain security refers to the application of industry chain theory to social and economic activities. Its connotation lies in the integration and optimization of the resources required for industrial development, including the purchase of raw materials, transportation, and capital supply. By exploring the influence of different links and elements in supply chain security, the influence elements of supply chain security on the wooden furniture industry and its internal mechanisms can be elucidated. The digital economy is a new form of economy in which information is digitized, data driven, and resource driven [48]. Research on digital economy theory is a practical exploration of industrial modernization and intellectualization and a vital link for the integration of digital technology with the real economy. Therefore, digital economy theory provides support and the basis for this study. In the era of the digital economy, data have become a new type of production element necessary for the industrial chain in each link. Meanwhile, the integration and innovation of open sharing brought about by digital technology make data an indispensable element for improving the supply chain security of the wooden furniture industry, thereby providing technical support and safeguarding supply chain security. It provides a theoretical foundation for the formulation of the research hypotheses and the construction of empirical models in this study.

4.2. Research Hypotheses

From the perspective of capital acquisition and utilization in the wood furniture industry, the development of the digital economy provides enterprises with richer financing channels, whereas real-time detection based on enterprise operating data can effectively improve the capital utilization rate [49]. This prompts the real economy to obtain financial gains in addition to goods sales, further ensuring the capital supply security of the wood furniture industry. Next, along with the increased popularity of digital technologies, such as mobile Internet, more accurate job search and recruitment matching effectively reduces the degree of information asymmetry between enterprises' labor demands and the labor market [50], ensuring labor supply to the industry. Moreover, the development of the digital economy integrates informatization, digitization, and intellectualization with the real economy [51], providing technological support for the sawing, cutting, chipping, planning, and even carving processes in the wooden furniture industry. Finally, regarding the raw materials supply of wooden furniture production, from the perspective of internal control, the development of digital technology can both provide enterprises with timely and accurate feedback production data to develop a reasonable production plan and provide richer procurement channels, standardized stocking processes, and accurate time schedules. Based on the external environment of the wooden furniture industry, the use of digital technology effectively simplifies the trading process and improves the efficiency of import customs clearance of raw materials, thus ensuring supply chain security for overseas supplies. Simultaneously, the development of digital technology can effectively improve the degree of informatization and modernization in forestry management, which in turn reduces risk in commercial forestry operations and guarantees raw materials supply security for the wooden furniture industry. Therefore, the following research hypothesis is proposed:

H1. The development of the digital economy has a significant positive impact on improving the level of supply chain security in China's wooden furniture industry.

Inventory turnover is a major indicator of an enterprise's supply chain performance [52]. The wooden furniture industry is a typical resource-intensive industry in which lightweight and precise inventory management is important for supply chain security. The development of the digital economy for the industry's inventory management ability plays an important and positive role in reducing the cost of supply information. The enterprise can screen all kinds of raw materials through the Internet and compare prices, consequently decreasing non-essential costs due to intermediaries. Moreover, the digital economy enables the industry to realize the intelligent management of the entire process of warehousing, replenishment, production, and delivery of raw materials using a digital warehouse management system, thus improving the enterprise's capital utilization efficiency and inventory management capability. Furthermore, digital technology enables enterprises to effectively engage with the wooden furniture market, accurately grasping the wooden furniture consumer market and identifying consumer demand to achieve customized production. This allows for the development of a more targeted and time-sensitive raw material procurement plan, effectively avoiding popular furniture being out-of-stock or off-sale and an excessive backlog in the remainder of the product, leading to funding fluctuations. Therefore, data mining, processing, analysis, and utilization of digital technology can effectively improve an enterprise's inventory turnover rate.

Inventory management is an essential factor in securing sustainable production in the wood furniture industry. Supply chain security includes not only labor supply and support for production technology but, more importantly, maintaining the supply of raw materials such as lumber, sawn timber, and panels needed for production, as well as capital chain security. Inventory turnover is a way for enterprises to convert products into capital and balance reserve funds and inventory volume, thus reducing operating costs. On the one hand, efficient inventory management effectively improves the short-term solvency of the enterprise and provides financial security. On the other hand, a high level of inventory turnover means a higher frequency of ordering, not only strengthening links with the suppliers of wood furniture enterprises but also modulating purchase decisions based on supplier prices in a timely manner. Additionally, efficient inventory management capability allows sufficient inventory space for the upcoming arrival of raw materials and avoids additional costs associated with expanding warehousing sites due to insufficient space.

The deep integration of digital technology into the real economy is an essential factor in expanding business boundaries. Wooden furniture enterprises can use digital technology to integrate and analyze the industry chain upstream and downstream key data information to formulate production, as well as the purchase of goods planning and decision-making to improve inventory management efficiency and effectiveness. Improving inventory management will further improve the solvency of wooden furniture enterprises and reduce operating costs, creating a predecessor for the supply chain security of the wooden furniture industry. Therefore, the following research hypothesis is proposed:

H2. The digital economy has a significant positive impact on supply chain security through improving the inventory management capability of the wooden furniture industry.

5. Research Design

5.1. Research Design

To deeply explore the impact of the digital economy on the supply chain security of China's wooden furniture industry and verify the research hypotheses, this study uses a two-way fixed effects model at the province and time levels for analysis. The regression equation is as follows:

$$supsec_{i,t} = \beta_0 + \beta_1 dig_{i,t} + \beta_k control_{i,t} + \gamma + \theta + \varepsilon_{i,t}$$
(1)

Equation (1) uses ordinary least squares (OLS) regression analysis, where $supsec_{i,t}$ denotes the supply chain security index of the wooden furniture industry for sample i in

year t and $dig_{i,t}$ denotes the digital economy development index of sample i in year t. The control are the control variables: β_k is the impact coefficient of the variables, γ and θ denote time and province fixed effects respectively, and $\varepsilon_{i,t}$ is the random error term.

Based on the above theoretical foundation, the inventory capacity of the wooden furniture industry is used as the mechanism variable, with reference to the classic literature, to construct the mediated effect regression equation [53] and explore the mechanism of the digital economy's influence on the industry's supply chain security. In addition, this study complements the Sobel and bootstrap tests as robustness tests after using the stepwise regression method as an initial test of the mediating effect. The specific regression equations are as follows:

$$sup_{i,t} = a_0 + a_1 dig_{i,t} + a_2 K_1 + \varepsilon_1 \tag{2}$$

$$sto_{i,t} = b_0 + b_1 dig_{i,t} + b_2 K_2 + \varepsilon_2 \tag{3}$$

$$sup_{i,t} = c_0 + c_1 dig_{i,t} + c_2 sto_{i,t} + c_3 K_2 + \varepsilon_3$$
(4)

where $dig_{i,t}$ indicates the digital economy development index of province (cities and districts) i in year t; $sup_{i,t}$ and $sto_{i,t}$ indicate the supply chain security index and inventory of the wooden furniture industry in province (cities and districts) i in year t, respectively; a_0 , b_0 , and c_0 are constant terms, K_1 and K_2 are control variables; and ε_i is a random disturbance term. Specifically, when the estimated coefficients a_1 , b_1 , and c_1 are all significant, this indicates that the development of the digital economy can improve the level of supply chain security by enhancing the inventory management capability of the wooden furniture industry. That is, the inventory optimization capability of furniture manufacturers has a mediating effect on the digital economy's impact on the supply chain security of the wooden furniture industry.

Finally, China is a major international timber-importing country, and maritime shipping is its most important import channel. Therefore, this study further explores whether the impact of the digital economy on the supply chain security of the wooden furniture industry varies according to geographic location by considering whether the sample provinces are in the coastal area as a geographic differentiation condition. The availability of port conditions in the sample provinces was also used as a supplement to the heterogeneity discussion to explore the impact of the digital economy on the supply chain security of China's wooden furniture industry in different samples. Moreover, to explore whether there is a difference in the impact of the digital economy on the supply chain security of the wooden furniture industry at different development stages and if this difference also exists in different geographic locations, this study draws on the results of existing research to construct a threshold effect model to carry out a more in-depth analysis. The specific regression equations are as follows:

$$sup_{i,t} = \begin{cases} \alpha_0 + \alpha_1 dig_{i,t} + controls + \varepsilon_{i,t}, dig \le \delta\\ \alpha_0 + \alpha_2 dig_{i,t} + controls + \varepsilon_{i,t}, dig > \delta \end{cases}$$
(5)

where subscripts i and t in Equation (5) represent the data corresponding to province and time, respectively. The model indicates that when the development index *dig* of the digital economy is located on the left side of the threshold δ , the impact coefficients of the digital economy on the supply chain security are α_1 and α_2 when it is on the right side.

It should be noted that a number of empirical tests were carried out to ensure the validity of the research results, as detailed below.

1. Multicollinearity test. Considering the large number of variables involved in the econometric analysis, the variance inflation factor (VIF) was used to avoid multi-collinearity that could lead to bias in the regression results. The test results showed that the maximum VIF value was 4.12. The mean value was 2.05, which is less than the strict VIF reference value of 5, indicating that there was no multicollinearity among the variables in the regression analysis.

- 2. Heteroskedasticity test. In order to avoid heteroskedasticity in the regression analysis that could reduce the explanatory validity of the impact of the digital economy's development on the supply chain security of China's wooden furniture industry, the regression model was subjected to the White test. The test results showed a *p*-value of 0.000, rejecting the original hypothesis of homoskedasticity at the 1% significance level. Consequently, robust standard errors were included in all subsequent empirical studies to address the problem of reduced explanatory validity due to heteroskedasticity.
- 3. Hausman test. The Hausman test was used to determine whether to choose a fixed effects model or a random effects model in the empirical analysis. The result of the test shows a *p*-value of 0.000, which rejects the original hypothesis of choosing a random-effects model at the 1% level of significance. Therefore, a fixed effects model was selected for the subsequent research.

5.2. Variable Selection

The dependent variable in this study is the wood furniture industry supply chain security index, and the independent variable is the digital economy development index. The mechanism variable is the inventory turnover ratio, which is an evaluation index reflecting the inventory management ability of the wood furniture industry and can effectively reflect enterprises' utilization efficiency in the production of raw materials. In this study, the selection of control variables is mainly considered from the following aspects. First, because of the differences in supply chain security requirements for wood furniture enterprises of different sizes, fixed assets as well as production factors in the furniture industry are controlled for. Second, given socioeconomic factors, the development of financial credit may affect the supply of funds in the wood furniture industry [54]; therefore, differences in the financial lending business development level across provinces (cities and districts) and supplementing the social employment rate as a reflection of the supply chain's labor supply were controlled for. Third, from the perspective of the consumer market for wooden furniture, population density differences directly affect the total demand for the product [55], which may affect the level of supply chain security. Fourth, forest cover was selected as a control variable for forest resource endowment, and road freight turnover was added to the control variables considering the impact of the transportation industry on industrial supply chain security [56].

5.3. Data Sources and Descriptive Statistics

The data used in this study were mainly obtained from the National Bureau of Statistics and the database of China's General Administration of Customs. The import and export data of wooden furniture, logs, sawn timber, artificial boards, wooden furniture processing, and production equipment involved in the calculation of the supply chain security index were all obtained from the database of China's General Administration of Customs, and the related trade commodities were screened using the HS 8-digit code. Statistics on the wooden furniture industry were obtained from the China Furniture Yearbook, China Light Industry Yearbook, and relevant reports published by the China National Furniture Association. Other macro data were mainly obtained from the China Statistical Yearbook, China Economic Census Yearbook, and provincial statistical yearbooks (cities and districts). In this study, 30 provinces (municipalities and autonomous regions in China, excluding Hong Kong, Macao, Taiwan, and Tibet) were selected as the research subjects from 2003 to 2021, and a total of 570 samples.

The selected variables and descriptive statistics are shown in Table 3. As can be seen, the development of China's wood furniture industry has large differences based on region or time. First, there is considerable variation in the industry's supply chain security level, with the lowest at 1.183 and the highest at 48.084. Similarly, the inventory turnover variable has a minimum value of 0.06 and a maximum value of 28.350, with a large standard deviation. Second, regarding the other variables affecting supply chain

security, the financial development level of each province (cities and districts) was relatively balanced, and the difference between the mean (1.271) and maximum value (2.759) was relatively small. Regarding forest cover, there was a large variation between the samples, with a maximum value of 66.8% and a minimum value of 2.94%, which is significantly different from the mean value of 31.156, indicating a large disparity in the forest resources of the provinces (cities and districts). Finally, regarding employment rate and transportation, the differences were small.

Name Variable Symbol Meaning Mean Std. N Index of supply chain Supply chain Dependent variable security in the wood 25.034 5.623 sup security index furniture industry Index of evaluation of Independent 3.678 1.458 Digital economy index dig digital economy variable development level Industry's operating Mediating 18.799 7.278 Inventory turnover ratio sto costs/Average balance of variable net inventories Log of the original cost of Original cost of fixed fixed assets in the assets in the furniture fas 2.381 2.184 furniture manufacturing manufacturing industry industry by province Log of annual wood Log of wood furniture furniture production 4.703 2.617 Inpro production by province Percentage of loan Development level of balances of financial financial and credit fina 1.271 0.447 institutions in GDP operations Control variables by province Log of total population Log of population density Inpden 5.441 1.314 density by province Percentage of forested area in each province in 17.698 Forest cover cove 31.156 relation to its administrative area Log of road Log of freight transported 8.761 0.684 frei freight turnover per kilometer of road Overall employment rates 0.704 Employment rate empl 96.523 by province

Table 3. Names and meanings of variables.

6. Results and Discussion

6.1. Benchmark Regression Results

To further analyze the impact of the digital economy development level of China's provinces (cities and districts) on the supply chain security of the wooden furniture industry, the results of different regression models were listed together. The regression results of Model 1 do not control for the effects of province and time. Model 2 only controls for time; Model 3 only controls for province; and Model 4 controls both the effects of time and region. The details are shown in Table 4.

Variables	1	2	3	4
Valiables	Sup	Sup	Sup	Sup
dia	2.725 ***	2.842 ***	1.881 ***	2.159 ***
dig	[0.210]	[0.316]	[0.247]	[0.362]
In Dood fusight turn over	1.178 ***	1.481 ***	1.670 ***	1.376 ***
ln Road freight turnover	[0.331]	[0.471]	[0.430]	[0.494]
In Wooden furniture production in 10,000 pieces	-0.1	0.002	0.083	0.063
In Wooden furniture production in 10,000 pieces	[0.149]	[0.138]	[0.136]	[0.137]
Einensiel development level	-1.227 ***	0.171	-0.051	0.859
Financial development level	[0.448]	[0.709]	[0.528]	[0.758]
Log of population density	-0.267	0.445	11.867 ***	13.110 ***
Log of population density	[0.165]	[0.470]	[2.380]	[2.404]
P	0.017 *	0.013	-0.037	0.017
Forest cover	[0.009]	[0.026]	[0.037]	[0.042]
Log of original cost of fixed accests	0.703 ***	0.158	0.066	0.231
Log of original cost of fixed assets	[0.197]	[0.184]	[0.177]	[0.186]
Employment rate	-1.698 ***	-0.663 **	-0.530 *	-0.404 *
Employment rate	[0.283]	[0.312]	[0.302]	[0.218]
Constantium	190.466 ***	88.838 ***	-12.514	-42.079
Constant term	[27.329]	[30.002]	[34.676]	[36.971]
Province fixed	No	No	Yes	Yes
Time fixed	No	Yes	No	Yes
Observed value	570	570	570	570
Fitted value	0.566	0.387	0.389	0.620

Table 4.	Empirical	test results.
----------	-----------	---------------

Note: Within [] are standard errors. *** p < 0.01, ** p < 0.05, * p < 0.1. Same as below.

According to the regression results, the impact of the digital economy on the supply chain security of China's wooden furniture industry passed the 1% significance level in all four regression results. Comparing the four regressions, the regression results showed a higher fit value of 0.620 when the two-way fixed-effects model was used, whereas the fit value was reduced to 0.387 in the model without two-way fixed effects. According to the regression results in Model 4, when controlling for other unchanged factors, for every 1 unit increase in the digital economy's development level, the supply chain security of the wood furniture industry increased by 2.159 units. Therefore, H1 is valid; that is, digital economy development has a significant positive impact on the supply chain security of the wooden furniture industry.

The regression results in the regression coefficients and significance analysis of the control variables also show that road freight turnover has a significant positive impact on the supply chain security of the wood furniture industry; that is, the transportation industry can effectively improve the supply chain security level of the wood furniture industry. The production of wooden furniture has a positive effect on the supply chain security level but does not pass the significance test. Financial industry development also passes significance only in the baseline regression, but the positive and negative regression coefficients differ depending on whether fixed effects are taken. However, when controlling for the time factor, the development of the financial industry has a negative impact on the supply chain security of the wooden furniture industry, whereas when the time factor is not controlled for, the impact is positive. This is likely because the wood furniture industry has a relatively long return on assets cycle, and the profit-seeking nature of capital makes it less willing to invest in such an industry with a lower long-term rate of return on assets [57]. Population density has a significant positive impact on supply chain security.

Areas with relatively high population density have a higher level of demand for wooden furniture and will also provide more adequate labor capacity resources for the industry, thus improving supply chain security. The effect of forest cover on supply chain security is positive, but this result does not pass the significance test. This may be due to the fact that raw materials for the production of wooden furniture in China rely more on imports, while policy implementation banning the logging of natural forests may further weaken the effect of forest cover on supply chain security [19]. There is a positive effect of total fixed assets in the furniture manufacturing industry on supply chain security; however, this result is only significant in the baseline regression and is no longer significant after controlling for year and province. The employment rate has a significant negative impact on the supply chain security of the wood furniture industry, which may be due to the fact that the wage level of jobs in the industry is relatively low, and thus the higher the social employment rate, the more difficult it is to seek labor resources.

6.2. Robustness Test

To further verify the positive impact of the digital economy on supply chain security in the wooden furniture industry, this study used three methods to test the robustness of the empirical results.

- Replacing the independent variable. The rapid development of the digital economy has benefited from investment in scientific and technological R&D by governments at all levels, as well as by all sectors of society. and a relatively higher investment in R&D will result in a higher level of economic development. Therefore, R&D intensity is used as a proxy variable for the digital economy to explore its impact on supply chain security in the wood furniture industry.
- 2. Replacing the dependent variable. Compared to the high-tech manufacturing industry, the wood furniture industry requires many laborers; thus, the number of existing laborers engaged in the wood furniture manufacturing industry also reflects the difference in the level of supply chain security of the industry to a certain extent. However, the employed population engaged in the industry does not affect the level of digital economy development in society, thus satisfying the fourth condition and avoiding the endogeneity problem to a certain extent. Therefore, this study replaces the supply chain security index with the existing number of employees engaged in the furniture manufacturing industry as the dependent variable to verify the impact of the digital economy on supply chain security.
- 3. Lagging 1–2 periods of the digital economy's level of development Considering that the wood furniture industry is a traditional manufacturing industry, the actual use of digital technology may lag behind the actual digital economy development, which leads to bias in the estimation results; thus, the digital economy development level of the lagged period is used as an instrumental variable in the regression analysis of the supply security of wood furniture. The results are summarized in Table 5.

	Replacing	Variables	Lagging 1–2 Period		
	Replacing the Independent Variable	Replacing the Dependent Variable	Lagging 1 Period	Lagging 2 Periods	
Variables	sup	lnlab	sup	sup	
dig		0.287 *** [0.097]	2.521 *** [0.336]	2.764 *** [0.328]	
rede	4.000 *** [0.537]				
Control variables	Control	Control	Control	Control	

Table 5. Robustness test.

	Replacing	Variables	Lagging 1–2 Period		
	Replacing the Independent Variable	Replacing the Dependent Variable	Lagging 1 Period	Lagging 2 Periods	
Time fixed	Yes	Yes	Yes	Yes	
Province fixed	Yes	Yes	Yes	Yes	
Sample size	570	567	540	510	
Goodness of fit	0.44	0.495	0.426	0.416	

Table 5. Cont.

Note: Within [] are standard errors. *** p < 0.01.

As can be seen, after replacing the independent and dependent variables, the positive impact of the digital economy on the supply chain security of the wooden furniture industry still passes the 1% significance level, which proves that the empirical results are robust. With a lag of 1 or 2 periods, the impact of the digital economy on the supply chain security of the wooden furniture industry remains significant, further verifying that the digital economy's positive impact on the supply chain security of China's wooden furniture industry is robust.

6.3. Impact Mechanisms Test

The benchmark regression analysis and robustness test results confirm the impact of the digital economy on supply chain security in China's wooden furniture industry. To further reveal the intrinsic mechanism of the impact of the digital economy on supply chain security, additional analyses are conducted. In the wooden furniture manufacturing industry, digital technology can effectively improve enterprises' production, supply, and marketing supervision and control. Through real-time data analysis and organization, the coordination and development of incoming and outgoing production and sales plans can clear the obstacles to sustainable operation. The inventory turnover rate is an important link between the use of digital technology and supply chain security for wooden furniture enterprises. Therefore, this study holds that the core logic lies in the following: The deep integration of the wooden furniture industry and the digital economy improves the industry's inventory management capability, which in turn provides prospective supply chain security. Details of the impact mechanism test are shown in Table 6.

Table 6. Impact mechanisms test.

	Basi	ic Regression Anal	ysis	Tv	vo-Way Fixed Effe	cts
Variables	1	2	3	4	5	6
	Sup	Sto	Sup	Sup	Sto	Sup
sto			5.585 ***			3.206 **
			[0.562]			[1.275]
dig	2.725 ***	0.189 ***	2.431 ***	2.159 ***	0.312 ***	1.280 **
Ū.	[0.210]	[0.015]	[0.299]	[0.362]	[0.012]	[0.538]
Control variables	Control	Control	Control	Control	Control	Control
Time fixed	No	No	No	Yes	Yes	Yes
Individual fixed	No	No	No	Yes	Yes	Yes
Sample size	570	570	570	570	570	570
R^2	0.566	0.525	0.631	0.420	0.722	0.427

Note: Within [] are standard errors. *** p < 0.01, ** p < 0.05.

The data in the table show that the coefficients a1, b1, and c1 pass the 1% significance level test regardless of whether the year or province are controlled for; that is, the inventory management capability of the wood furniture industry plays a mediating role in the impact of the digital economy on supply chain security. After controlling for time and individual

effects, the extent to which the digital economy influences the improvement in inventory management capability in the wood furniture industry was significantly higher than that in the benchmark regression. In the benchmark regression, for every 1 unit increase in the digital economy development level, the supply chain security level increased by 2.725 units and the inventory turnover ratio increased by 0.189. Whereas, in a two-way fixed effects regression, for every 1 unit increase in the digital economy development level, the supply chain security level increased by 2.159 units and the inventory turnover ratio increased by 0.312, which means that the digital economy has a significant positive impact on the inventory management capability of the wooden furniture industry and improves supply chain security through inventory management.

To ensure the robustness of these empirical results, the Sobel and bootstrap tests were used for further validation. According to the research of mediating effects in the existing literature, the Sobel test is more persuasive than the traditional three-step test [58], which can directly calculate the proportion of the mediating effects. Depending on the statistic $z = \hat{a}\hat{b}/s_{ab}$, where \hat{a} and \hat{b} are the estimates of a and b, respectively, the standard error is: $se(ab) = \sqrt{\hat{a}^2 se_b^2 + \hat{b}^2 se_b^2}$. In contrast, the bootstrap test takes a sample with a larger sample size as a whole and draws from it with putbacks to obtain a more accurate standard error [55]. Therefore, the test was conducted with 500 times of sampling with replacement in 30 provinces (cities and districts). The test results are shown in Table 7.

Table 7. The Sobel test and bootstrap test.

			Sobe	l Test			Bootstr	ap Test
Statistics	Sobel	Coefficient a	Coefficient b	Mediating Effect	Direct Effect	Total Effect	Indirect Effect	Direct Effect
Coefficient	0.999	0.312	3.206	0.999	1.160	2.159	0.999	1.160
Std.	0.399	0.012	1.275	0.399	0.536	0.362	0.354	0.565
Z-value	2.503	25.018	2.515	2.503	2.165	5.969	2.82	2.050
<i>p</i> -value	0.012	0.000	0.012	0.012	0.030	0.000	0.005	0.040

As shown in Table 8, the z-value in Sobel's test was 2.503, with a *p*-value of 0.012. In the bootstrap test, the z-value was 2.82 and the *p*-value was 0.005. Therefore, both tests rejected the original hypothesis of H0: ab = 0 at the 5% significance level. The impact of the digital economy on the supply chain security of the wooden furniture industry is mediated by the inventory turnover rate, accounting for 46.27%; in essence, almost 46% of the digital economy's influence on supply chain security is attained through the enhancement of inventory management capabilities.

Table 8. Regional heterogeneity tests.

Variables	All-Sample	Coastal Province	Non-Coastal Province	With Ports	Without Ports
variables =	Sup	Sup	Sup	Sup	Sup
dig	2.159 ***	0.254	4.601 ***	0.140	3.231 ***
0	0.362	0.539	0.995	0.473	0.419
Control variables	Control	Control	Control	Control	Control
Time fixed	Yes	Yes	Yes	Yes	Yes
Province fixed	Yes	Yes	Yes	Yes	Yes
Sample size	570	570	570	570	570
Goodness of fit	0.420	0.381	0.712	0.316	0.682
<i>p</i> -value	/	0.	000 ***	0.	026 **

Note: *** p < 0.01, ** p < 0.05; *p*-values for coefficient differences are based on estimates from the Chow test of the interaction model. In addition, the results based on the likelihood of no correlation test and the Fisher's combination test are in general agreement.

6.4. Further Study

Raw materials supply is vital for supply chain security in the wood furniture industry. Combined with the current development of China's forestry industry, the raw material supply for many forest products still relies on overseas imports [59]. UN trade statistics show that China's dependence on timber imports has remained above 50% over the past ten years, and the availability of ports of entry for foreign trade has become an important factor in the supply chain security of the wooden furniture industry. Generally, compared with inland provinces, coastal provinces have a higher level of economic development because of their geographic location, and the degree of integration of digital technology with the real economy is increased accordingly. Shipping is a major import channel for primary raw materials, such as logs and sawn timber, thus inferring that there may be regional differences in the impact of the digital economy on the supply chain security of China's wooden furniture industry. Therefore, a more in-depth analysis from economic and geographical perspectives is necessary. Together with the development conditions of the wooden furniture industry, this study takes the sample province "Whether coastal or not" 1 as the grouping condition to do further research, and the sample province "Whether port or not" 2 was taken as a supplement. The Chow test was conducted as the robustness of heterogeneity test. The test results are shown in Table 8.

Based on the test results, the impact of the digital economy on the supply chain security of the wooden furniture industry in coastal areas was non-existent, while in non-coastal areas or samples without ports, it had a significant impact. Table 9 shows that when controlling for other factors that remain unchanged in non-coastal provinces, every 1 unit increase in the digital economy increases the supply chain security index of the wood furniture industry by 4.601. In provinces without ports, for every 1 unit increase in the digital economy's development level, the supply chain security level of the wood furniture industry increases by 3.231. Furthermore, the Chow test results show that the *p*-values of the differences in the coefficients of the grouped interaction terms all reject the original hypothesis that the residuals ε are independently and identically normally distributed with unknown variance at the 5% significance level.

		Whether Coastal or Not			Port or Not
	All-Sample	Yes	No	With	Without
Variables	sup	sup	sup	sup	sup
heta	1.825 **	5.205	1.749 ***	5.533	1.748 **
$\operatorname{dig} \leq \theta$	4.547 ***	2.414 **	6.489 ***	2.542 ***	5.717 ***
0 =	[1.026]	[0.702]	[1.136]	[0.559]	[1.247]
dig > θ	1.866 ***	1.789 **	2.396 ***	1.952 ***	1.960 **
0	[0.398]	[0.722]	[0.526]	[0.460]	[0.709]
95% confidence interval	[1.724, 1.897]	[5.0941, 5.3211]	[1.580, 1.760]	[5.415, 5.651]	[1.566, 1.760]
Control variables	Control	Control	Control	Control	Control
Province fixed	Yes	Yes	Yes	Yes	Yes
Time fixed	Yes	Yes	Yes	Yes	Yes
R ²	0.458	0.658	0.425	0.650	0.368

Table 9. The threshold regression test.

Note: Within [] are standard errors. *** p < 0.01, ** p < 0.05.

However, considering that the test results fall short of intuitive expectations, one possible reason is that, at this stage, digital technology has not yet made new breakthroughs, and technological tools applied to the wooden furniture industry have a limited role in improving supply chain security [60]. This leads to the nonlinear impact of the digital economy on the supply chain security of the wooden furniture industry. When the digital economy development level exceeds a certain degree, the positive promotion effect on the supply chain security of the wooden furniture industry is weakened. Therefore, a threshold

regression model was used to further explore the reasons for this heterogeneity. The test results are shown in Table 9.

As Table 9 demonstrates, in the overall sample, the coefficient of impact on supply chain security in China's wooden furniture industry was 4.547 when the digital economy development index is on the left side of the threshold of 1.825, and 1.866 when it is on the right side of the threshold; both pass the test at the 1% level of significance. Thus, it is evident that the digital economy's impact on the wood furniture industry's supply chain security has a threshold effect. This impact does exist in the case of a marginal diminution of the development level of the digital economy exceeding the threshold value, and the impact on supply chain security weakens. Moreover, the results of the "Whether coastal or not" and "Whether port or not" subgroups show that the threshold effect exists only in the non-coastal and non-port samples, with thresholds of 1.749 and 1.748, respectively, similar to the results in the full sample. For coastal provinces and provinces with ports, although the threshold value fails the significance test, the impact of the digital economy on the supply chain security of the wooden furniture industry on both sides of the threshold value shows the same change, from high to low. Specifically, in the non-coastal sample, the coefficient of impact on supply chain security was 2.396 when the digital economy index was on the right side of the threshold and improved to 6.489 when it was on the left side, both passing the 1% significance test. For samples without ports, the impact coefficient was 1.960 when the digital economy index was on the right side of the threshold and 5.717 when it was on the left; the test results passed the 5% significance test.

Clearly, the impact of the digital economy on the supply chain security of China's wooden furniture industry varied significantly under different geographic location conditions, further validating the heterogeneity test results. Moreover, the threshold effect test results show that the impact of the digital economy on the supply chain security of the wood furniture industry has a marginally decreasing trend, and this trend also exists in regional heterogeneity.

7. Conclusions and Recommendations

This study explored the impact of the digital economy on supply chain security by constructing an evaluation index system of the digital economy and supply chain security of the wooden furniture industry and reveals its intrinsic impact mechanism by constructing a mediating effect model. Further analysis revealed regional heterogeneity in the digital economy's impact, and the intrinsic reasons for the differences in heterogeneity are explored by constructing a threshold effect model. The following conclusions were drawn from the empirical test results of this study:

First, the digital economy can improve the supply chain security of China's wooden furniture industry. Second, inventory management capability is an essential link to the digital economy's impact. Inventory management capability guarantees the smooth coordination of China's wooden furniture industry chain upstream and downstream operations. Digital economy development accelerates the integration of traditional manufacturing and digital technology through platform integration, information sharing, data analysis, and other digital methods to promote the digital transformation of the wooden furniture industry and improve the inventory turnover rate. This is not only for back-end production to provide protection but also for the front-end purchase of goods to provide space for their preparation. Third, there is heterogeneity in the digital economy's impact on supply chain security. The digital economy has a significant positive impact on the non-coastal and nonport regions, whereas it is not significant in the coastal and port provinces. Fourth, there is a threshold effect on the impact of the digital economy on supply chain security in the wood furniture industry; the positive impact is significantly higher when the digital economy development level is on the left side of the threshold than on the right side. Compared to its modern high-tech counterpart, the wooden furniture industry, as a typical representative of the traditional manufacturing industry, has a relatively short supply chain, and the degree of integration and embeddedness of digital technology is limited, which does not achieve

sustained growth in improving the supply chain security of China's wooden furniture industry. Therefore, the enhancement of supply chain security is more effective in inland regions where the digital economy has a relatively low level of development, whereas the digital economy plays a more limited role in southeastern coastal regions.

Considering the findings of this study, the following policy recommendations are suggested. From the perspective of digital economic development, the intensity and breadth of digital infrastructure development should be increased. First, to accelerate the transformation of new and old kinetic energy between the digital economy and traditional manufacturing industries, improving the level of integration and embeddedness between the two increases investment in key digitalization research. This enhances innovation and creativity, laying the foundation for upgrading the basic industries of the digital economy and enhancing the core competitiveness of the digital technology industry. Second, to further expand the influence of the digital economy on the wood furniture industry and constantly expand its integration into traditional manufacturing and embeddedness from the coast to the inland, a deep integration of digital and traditional manufacturing industries should be undertaken to reduce the status quo of China's unbalanced regional development in the wood furniture industry.

From the perspective of supply chain security in China's wooden furniture industry, extending the length of the supply chain and seeking new combinations and growth points for the supply chain and modern technology are important. First, to further improve the degree of standardization and programming, of the various links in the supply chain and provide a more adequate basis for programmatic and intelligent embedding of digital technology. By introducing advanced information technology, such as ERP systems, production processes can be effectively managed, and efficiency and accuracy can be improved. Second, to improve the utilization rate of platforms and data in the wooden furniture industry, in terms of wood furniture market identification, big data and cloud computing technology can be used to analyze market demand and consumer behavior so as to guide product design and production. For example, by analyzing historical sales data, it is possible to predict future market trends and optimize inventory management and production planning. This data can help expand the supply channels of raw materials for the production of wooden furniture and seek the optimal supplier by combining information on consumer demand and operating costs. These changes improve market competitiveness while further compressing operating costs, increasing the digital economy's impact on the wooden furniture industry's ability to perform better.

Author Contributions: Conceptualization, Y.L. and F.L.; methodology, Y.C. and F.L.; formal analysis, C.Y. and F.X.; data curation, C.T.; writing—original draft preparation, Y.L. and F.L.; writing—review and editing, C.X. and C.Y. All authors have read and agreed to the published version of the manuscript.

Funding: National Natural Science Foundation of China Youth Project "Study on the process, mechanism and policy regulation of forest resources-industry decoupling" (72003013) and the basic scientific research business fee project of central universities "Forestry carbon sink value diversification realization way research" (2023SKY20).

Data Availability Statement: Data are available upon request.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- 1. Fridell, G. The political economy of inclusion and exclusion: State, labour and the costs of supply chain integration in the Eastern Caribbean. *Rev. Int. Political Econ.* **2022**, *29*, 749–767. [CrossRef]
- Lin, M.; Lu, H.; Sun, J. The adjustment trend of global supply chain pattern and China's coping strategies. *Int. Trade* 2020, 10, 19–25.
- 3. Liu, Q.; Ning, Z. Impact of Global Supply Chain Crisis on Chinese Forest Product Enterprises: Trade Trends and Literature Review. *Forests* 2023, 14, 1247. [CrossRef]
- 4. Izmailova, M.A. The impact of digital economy on the transformation of the labor market and forming new business models. *Russ. J. Ind. Econ.* **2018**, *11*, 296–304. [CrossRef]

- 5. Bai, M. Ten new import concepts of China's trade development. *Int. Econ. Coop.* **2020**, *6*, 81–86.
- 6. Wei, T.; Tian, M.; Ma, S.; Wang, F. Analysis of the substitutability of China's timber imports and the safety of import sources. *For. Econ. Issues* **2021**, *41*, 172–179.
- Ge, H.P.; Wu, F.X. Digital economy empowers high-quality economic development: Theoretical mechanism and empirical evidence. *Nanjing Soc. Sci.* 2021, 17, 24–33.
- Clarkson, G.; Jacobsen, T.E.; Batcheller, A.L. Information asymmetry and information sharing. *Gov. Inf. Q.* 2007, 24, 827–839. [CrossRef]
- 9. Steyaert, J.; Gould, N. Social work and the changing face of the digital divide. Br. J. Soc. Work. 2009, 39, 740–753. [CrossRef]
- 10. Vial, G. Understanding digital transformation: A review and a research agenda. *Manag. Digit. Transform.* **2019**, *28*, 13–66. [CrossRef]
- Jing, W.J.; Sun, B.W. Digital economy promotes high-quality economic development: A theoretical analysis framework. *Economist* 2019, 2, 66–73.
- Tan, L.S.; Yang, Z.D.; Muhammad, L.; Ding, C.J.; Hu, M.J.; Hu, J. Toward low-carbon sustainable development: Exploring the impact of digital economy development and industrial restructuring. *Bus. Strategy Environ.* 2023, 33, 2159–2172. [CrossRef]
- 13. Eyvazzadeh, E.; Khamseh, A.; Khavar, B.C. Investigating the Role of Collaborative Innovation Networks and Customer Participation on New Product Performance Cinnagen Co. *Revista Geintec-Gestao Inovacao Tecnologias* **2021**, *11*, 911–922. [CrossRef]
- 14. Sandra, M.; Reem, T.; Chaza, A. The impact of digital technology on health of populations affected by humanitarian crises: Recent innovations and current gaps. *J. Public Health Policy* **2016**, *37*, 167–200. [CrossRef] [PubMed]
- 15. Tatarinov, A.A. Measuring digital economy in national accounts. Vopr. Stat. 2019, 26, 5–17. [CrossRef]
- 16. Miller, Y.V. Creation of added value in the context of measurement complexity of the digital economy. *E-Management* **2020**, *3*, 68–74. [CrossRef]
- 17. Han, D.; Ding, Y.; Shi, Z.; He, Y. The impact of digital economy on total factor carbon productivity: The threshold effect of technology accumulation. *Environ. Sci. Pollut. Res.* 2022, 29, 55691–55706. [CrossRef] [PubMed]
- Xu, A.; Qian, F.; Pai, C.H.; Na, Y.; Pan, Z. The impact of COVID-19 epidemic on the development of the digital economy of China—Based on the data of 31 provinces in China. *Front. Public Health* 2022, *9*, 778671. [CrossRef] [PubMed]
- Serna Gómez, J.H.; Díaz-Piraquive, F.N.; Muriel-Perea, Y.D.J.; Díaz Peláez, A. Advances, opportunities, and challenges in the digital transformation of HEIS in Latin America. In *Radical Solutions for Digital Transformation in Latin American Universities: Artificial Intelligence and Technology 4.0 in Higher Education*; Springer: Berlin/Heidelberg, Germany, 2021; pp. 55–75. [CrossRef]
- 20. Achuora, J.; Odoyo, J.; Kenyatta; Odoyo, A.J. A Review and Research Direction of Green Supply Chain Management in Kenya. 2018. Available online: https://www.researchgate.net/publication/343821250 (accessed on 8 April 2024).
- 21. Sheng, C.X. The ideas and strategies to promote the safe and stable development of the industrial chain supply chain under the new development pattern. *Reformation* **2021**, *2*, 1–13.
- 22. Dong, Q. Research on MNCs' Supply Chain Implementation in China. Contents, Problems and Recommendations. Ph.D. Thesis, University of Grenoble, Grenoble, France, 2011.
- 23. Li, T.J.; Zhao, X.J. Exploration of securing industrial chain supply chain in new China. Manag. World 2022, 9, 31–41.
- 24. Zhang, Z.B.; Wang, X.K. Chinese-style modernization: Theoretical foundation, ideological evolution and practical logic. *Adm. Reform.* **2021**, *8*, 4–12.
- 25. Yang, X.B. Applied-information Technology in Supply Chain Knowledge Collaborative Model based on Semantic Web Service. *Adv. Mater. Res.* **2014**, *908*, 535–538. [CrossRef]
- 26. Bertani, F.; Ponta, L.; Raberto, M.; Teglio, A.; Cincotti, S. The complexity of the intangible digital economy: An agent-based model. *J. Bus. Res.* **2021**, *129*, 527–540. [CrossRef]
- Yang, J.J.; Ai, W.W.; Fan, Z.J. Scenarios, governance and response to the digital economy-enabled global industrial chain supply chain division of labor. *Economist* 2022, 9, 49–58.
- 28. Hu, C.; Li, Y.; Zheng, X. Data assets, information uses, and operational efficiency. Appl. Econ. 2022, 54, 6887–6900. [CrossRef]
- 29. Liu, D.X.; Chen, H. Product-service supply chain pricing decision: Impact analysis of data resource mining and sharing strategy. *Chin. J. Manag. Sci.* **2023**, *32*, 1–15.
- 30. Peng, Z.Q.; Zhou, P. Research on the impact of digital transformation on commercial credit financing ability—Based on the perspective of supply chain information transfer. *Mod. Manag.* **2023**, *1*, 82–90.
- Kulyasova, E.V.; Trifonov, P.V. Development of forms of interaction between universities and the business community in the digital economy. *Strateg. Decis. Risk Manag.* 2020, 11, 216–223. [CrossRef]
- Crawford, J.; Lindvall, J. Leveraging digital technologies in Enterprise Risk Management. Manag. Inf. Technol. Digit. Transform. 2021, 3, 159–171. [CrossRef]
- Mao, Z.G.; Zhou, L.X.; Chen, C.Q. Study on the Countermeasures for the Transformation and Upgrading of Wood Industry in Jiangshan City—Reflections on the Enlightenment of the Cluster Development of Furniture Industry in Nankang District of Ganzhou City. *Green China* 2022, 10, 64–68.
- Lin, M. The Conflict between Technology and Scale: Evidence from China's Wooden Furniture Industry. Sustainability 2023, 15, 230. [CrossRef]
- 35. Lee, J.H.; Na, D.S.; Jung, J.T. A study on the impact of intangible assets on corporate value. *Indian J. Public Health Res. Dev.* 2018, 9, 422. [CrossRef]

- 36. Cosmulese, C.G.; Socoliuc, M.; Ciubotariu, M.S.; Grosu, V.; Dorel, M. Empirical study on the impact of evaluation of intangible assets on the market value of the listed companies. *Ekon. Manag.* **2021**, *1*, 84–101. [CrossRef]
- 37. Sardo, F.; Serrasqueiro, Z. Intellectual capital and high-tech firms' financing choices in the European context: A panel data analysis. *Quant. Financ. Econ.* **2021**, *5*, 1–18. [CrossRef]
- Feng, Z.; Zhao, K. Employment-based health insurance and aggregate labor supply. J. Econ. Behav. Organ. 2018, 154, 156–174. [CrossRef]
- 39. Wiertz, D.; Lim, C. The civic footprints of labor market participation: Longitudinal evidence from the United States, 2002–2015. *Soc. Forces* **2019**, *97*, 1757–1784. [CrossRef]
- 40. Rosa, M.D.; Gainsford, K.; Pallonetto, F. Diversification, concentration and renewability of the energy supply in the European Union. *Energy* **2022**, *253*, 124097. [CrossRef]
- 41. Ke, J.Y.F.; Shabbir, T.; Corona, J. The impact of exchange rate volatility on the industry-level geographic diversification of global supply chain network. *Int. J. Logist. Econ. Glob.* **2018**, *7*, 366–387. [CrossRef]
- 42. Nouri, K.; Abdul-Nour, G. Optimization via Computer Simulation of a Mixed Assembly Line of Wooden Furniture-A Case Study. *Procedia Manuf.* **2019**, *39*, 956–963. [CrossRef]
- 43. Li, H.J.; Cheng, B.D.; Yang, J. Research on the impact of epidemic on the layout of global value chain of timber industry—Based on the general equilibrium model of global value chain. J. Agrotech. Econ. 2022, 6, 81–98.
- 44. Ambrušová, L.; Šulek, R. Factors influencing forest owners and manager's decision making about forestry services in loggingtransport process/Faktory vplývajúce na rozhodnutia vlastníkov a obhospodarovateľov lesov o spôsobe zabezpečovania lesníckych služieb vťažbovo-dopravnom výrobnom procese. *Cent. Eur. For. J.* 2014, 60, 177–184. [CrossRef]
- 45. Tang, M.; Liu, Y.; Ding, F.; Wang, Z. Solution to solid wood board cutting stock problem. Appl. Sci. 2021, 11, 7790. [CrossRef]
- 46. Bai, P.W.; Zhang, Y. The digital economy, declining demographic dividend, and the rights of middle- and low-skilled workers. *Econ. Res.* **2021**, *5*, 91–108.
- 47. Qu, Y.Y. Theoretical connotation of industrial chain length and its function realization. China Ind. Econ. 2022, 5–24.
- 48. Shpak, P.S.; Sycheva, E.G.; Merinskaya, E.E. The concept of digital twins as a modern trend of the digital economy. *Bull. Omsk. Univ. Ser. Econ.* **2020**, *18*, 57–68. [CrossRef]
- Yuan, S.; Musibau, H.O.; Genç, S.Y.; Shaheen, R.; Ameen, A.; Tan, Z. Digitalization of economy is the key factor behind fourth industrial revolution: How G7 countries are overcoming with the financing issues? *Technol. Forecast. Soc. Chang.* 2021, 165, 120533. [CrossRef]
- 50. Srinidhi, V.; Karachiwala, B.; Iyer, A.; Reddy, B.; Mathrani, V.; Madhiwalla, N.; Sen, G. ASHA Kirana: When digital technology empowered front-line health workers. *BMJ Glob. Health* **2021**, *6* (Suppl. 5), e005039. [CrossRef] [PubMed]
- 51. Liu, H.; Li, S. The Impact of Urban Digital Economy Development on Manufacturing Innovation Efficiency: Evidence from Chinese Listed Manufacturing Firms. *Int. J. Empir. Econ.* **2023**, *2*, 2350004. [CrossRef]
- 52. Rao, C.M.; Rao, K.P. Inventory turnover ratio as a supply chain performance measure. *Serbian J. Manag.* **2009**, *4*, 41–50.
- 53. Wen, Z.; Chang, L.; Hau, K.T.; Liu, H. Testing and application of the mediating effects. Acta Psychol. Sin. 2004, 36, 614–620.
- 54. Cuéllar Sánchez, D.; Dueñas Peña, A.; Núñez-Valdés, K. The impact of credit on small regional enterprises: A multidisciplinary observational study. *Russ. Law J.* 2023, *11*, 655–663. [CrossRef]
- 55. O'Hara, J.K.; Lin, Z.Q. Population density and local food market channels. *Appl. Econ. Perspect. Policy* 2020, 42, 477–496. [CrossRef]
- 56. Wang, H.; Han, J.Y.; Su, M.; Wan, S.L.; Zhang, Z.C. The relationship between freight transport and economic development: A case study of China. *Res. Transp. Econ.* 2020, *85*, 100885. [CrossRef]
- 57. Nwankwo, C.H.; Igweze, A.H. Comparison of Tests of Indirect Effect in Single Mediation Analysis. *Am. J. Theor. Appl. Stat.* 2016, 5, 64–69. [CrossRef]
- 58. Zhu, Y.L.; Sun, Y.N.; Xiang, X.Y. 'Profit' or 'Growth'? Research on the correlation between capital structure and enterprise value. *China Econ. Issues* **2019**, *6*, 104–118.
- Wang, D.D. Risks and Avoidance of China's Timber Import Overseas Transportation—Comment on 'Research on the Characteristics, Elasticity and Risks of China's Timber Import Market'. For. Econ. 2020, 42.
- 60. Zhu, J.G.; Wang, X. Analysis of intelligent manufacturing enabling technology and development path of wood furniture. *J. For. Eng.* **2021**, *6*, 177–183.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.